Author’s response to reviews

Title: A 95 year-old suffering circulatory arrest after accidental hypothermia: a case report

Authors:
Anders Wetting Carlsen (anders.wetting.carlsen@stolav.no)
Anders Marvik Winnerkvist (anders.marvik.winnerkvist@stolav.no)
Guri Greiff (guri.greiff@stolav.no; guri.greiff@gmail.com)

Version: 2 Date: 11 Oct 2017

Author’s response to reviews:

Dear Editor. Once again thank you for Your revision. The following points have been changed in the manuscript (according to Your suggestions):
1. The abstract entered into Editorial Manager is now identical with the one in the manuscript.
2. Clarification of what the patient’s next of kin have consented to being published, is added under "Declarations-Consent for publication", line 154-155.
3. and 4. Changed in accordance to Your suggestion, see line 165-168.